Author’s response to reviews

Title: Occurrence of colorectal aberrant crypt foci depending on age and dietary patterns of patients

Authors:

Marek Kowalczyk (forel@neostrada.pl)
Marcin Orłowski (m.orlowski@przychodniachelm.pl)
Piotr Siermontowski (nurgok@tlen.pl)
Dariusz Mucha (nauka.autograf@gmail.com)
Krzysztof Zinkiewicz (kzinek@yahoo.com)
Waldemar Kurpiewski (wkurpiewski@interia.pl)
Ewa Zieliński (ewa.zielinski.bydgoszcz@wp.pl)
Iwona Kowalczyk (iwonakowalczyk71@wp.pl)
Agnieszka Pedrycz (apw4@wp.pl)

Version: 2 Date: 27 Dec 2017

Author’s response to reviews:

Response for the reviewer Proffesor Luca Roncucci

Ad 1

Dear Professor, we would like to inform you that our article has been repeatedly tested in the field of English. Language adjustments were made by translators cooperating with the publishing house “Undersea & Hyperbaric Medicine Society”. The translators were a native Englishman and a native American.

Dear Professor, if our article requires a new linguistic correction, we ask you to indicate the pages and lines that require this correction.
Ad 2

Dear Professor, relevant statistical data is contained in the text of the article, with descriptions of figures on page 6, lines from 1 to 55.

Ad 3

Dear Professor, we have introduced the appropriate correction on page 5, line 49, now the number of the bibliographic reference is (28).